Author’s response to reviews

Title: Propensity score matching analysis of a phase II study on simultaneous modulated accelerated radiation therapy using helical tomotherapy technique in patients with nasopharyngeal carcinoma

Authors:
Lei Du (coolbile_1999@163.com)
Xinxin Zhang (xinxinzhang66@hotmail.com)
Linchun Feng (301flc@163.com)
Baolin Qu (qubl6212@sina.com)
Jing Chen (chenjing19861206@163.com)
Jun Yang (yang74008@163.com)
Haixia Liu (crystal0214@163.com)
Shouping Xu (xshp228@163.com)
Chuanbin Xie (xiechuanbin2003@163.com)
Lin Ma (malinpharm@sina.com)

Version: 5 Date: 19 Aug 2017

Author’s response to reviews:

Editor Comments:

1. As your article reports the results of a health care intervention on human participants, it must be registered in an appropriate registry and the registration number and date of registration should be stated in the Abstract section (as per our submission guidelines: https://bmccancer.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article). If it was not registered prospectively (before enrollment of the first participant), you should include the words 'retrospectively registered'.

Response: The detail of ‘Trial registration’ has been added in the Abstract section.